Author's response to reviews

Title: Associations between proinflammatory cytokines in the synovial fluid and radiographic grading and pain-related scores in 47 consecutive patients with osteoarthritis of the knee

Authors:

Sumihisa Orita (sumihisa@silver.email.ne.jp)
Takana Koshi (mocodo5@yahoo.co.jp)
Takeshi Mitsuka (mituka.m@fc-h.jp)
Masayuki Miyagi (masayuki008@aol.com)
Gen Inoue (ginoue@faculty.chiba-u.jp)
Gen Arai (gen.arai.tateyama@hotmail.co.jp)
Tetsuhiro Ishikawa (ishikawa_tetsuhiro@yahoo.co.jp)
Eiji Hanaoka (eihana@aol.com)
Keishi Yamashita (kctha0311.---@ezweb.ne.jp)
Masaomi Yamashita (masaomiym@aol.com)
Yawara Eguchi (yawara_eguchi@yahoo.co.jp)
Tomoaki Toyone (toyone@med.teikyo-u.ac.jp)
Kazuhisa Takahashi (19501114@faculty.chiba-u.jp)
Seiji Ohtori (sohtori@faculty.chiba-u.jp)

Version: 10 Date: 29 June 2011

Author's response to reviews:

We deleted line numbers, hilights, strikethrough, and underscores. Also we deleted some parts which should be done after the major revision.